

**Author's response to reviews**

**Title:** Delayed posttraumatic spinal cord infarction in an adult following minor head injury: a case report

**Authors:**

Viktor Bartanusz (bartanusz@uthscsa.edu)
Mateo Ziu (ziu@uthscsa.edu)
Leisha E Wood (leisha_ewood@yahoo.com)
Jean-Louis Caron (caron@uthscsa.edu)

**Version:** 3 **Date:** 19 April 2012

**Author's response to reviews:**

Professor Michael Kidd AM
Editor-in-Chief
Flinders University, Australia

April 19, 2012, San Antonio

Dear Editor,

Please find enclosed the revised version of our manuscript “Delayed posttraumatic spinal cord infarction in an adult following minor head injury: a case report”.

We attentively followed all your recommendations and made the following changes:

1. Added the ethnicity of the patient
2. Subdivided the Case presentation section to appropriate subheadings
3. Provided detailed steps taken to obtain the consent from the patient’s family
4. Made sure that the patient’s gender was appropriate throughout the manuscript

Thank you for your consideration.

Sincerely,

Viktor Bartanusz, MD, PhD
Department of Neurosurgery
University of Texas Health Science Center at San Antonio
7703 Floyd Curl drive, San Antonio, Texas 78229-3900
Tel.: + 1 210 567 5625
Fax: +1 210 567 6066
E-mail: bartanusz@uthscsa.edu